Author’s response to reviews

Title: A heterozygous duplication variant of the HOXD13 gene caused synpolydactyly type 1 with variable expressivity in a Chinese family

Authors:

Tahir Zaib (zaibi_hb@yahoo.com)
Wei Ji (jiwei@ems.hrbmu.edu.cn)
Komal Saleem (komalsaleem93@yahoo.com)
Guangchen Nie (surgeon626@163.com)
Chao Li (hmu0272@126.com)
Lin Cao (372321402@qq.com)
Baijun Xu (xbjmhgf@163.com)
Kexian Dong (dongkexian@hrbmu.edu.cn)
Hanfei Yu (yhf@hrbmu.edu.cn)
Xuguang Hao (haoxg123@163.com)
Yan Xue (xueyan1188@163.com)
Shuhan Si (1820410367@qq.com)
Xueyuan Jia (jiaxueyuan@hrbmu.edu.cn)
Jie Wu (wujie@hrbmu.edu.cn)
Xuelong Zhang (zhangxuelong@hrbmu.edu.cn)
Rongwei Guan (guanrongwei@hrbmu.edu.cn)
Guohua Ji (jiguohua@hrbmu.edu.cn)
Jing Bai (baijingbj@hrbmu.edu.cn)
Feng Chen (chenfeng@hrbmu.edu.cn)
Yong Liu (uceduc@163.com)
Wenjing Sun (sunwj@ems.hrbmu.edu.cn)
Songbin Fu (fusb@ems.hrbmu.edu.cn)
Version: 3 Date: 10 Oct 2019

Author’s response to reviews:

Thanks a lot for the editor, you have made the essential revisions on the manuscript. Thank you very much for improving our manuscript a lot. We have resubmitted the clean version in the system. Thank you for your consideration of this manuscript for BMC Medical Genetics.